Author's response to reviews

Title: Use of an automated blood culture system (BD BACTECTM) for diagnosis of prosthetic joint infections: easy and fast

Authors:

   Angela M Minassian (angela.minassian@ouh.nhs.uk)
   Robert Newnham (robert.newnham@ouh.nhs.uk)
   Elizabeth Kalimeris (elizabeth.kalimeris@ouh.nhs.uk)
   Philip Bejon (pbejon@well.ox.ac.uk)
   Bridget L Atkins (bridget.atkins@ouh.nhs.uk)
   Ian CJW Bowler (ian.bowler@ouh.nhs.uk)

Version: 4  Date: 18 April 2014

Author's response to reviews: see over
Dear Editors

Re: Use of an automated blood culture system (BD BACTEC™) for diagnosis of prosthetic joint infections: easy and fast

1,2Angela M Minassian, 1Robert Newnham, 1Elizabeth Kalimeris, 1,2Philip Bejon, 1,2Bridget L Atkins, 1Ian CJW Bowler

1Department of Microbiology, Oxford University Hospitals NHS Trust, John Radcliffe Hospital, Headley Way, Headington, Oxford, OX3 9DU, UK; 2Bone Infection Unit, Nuffield Orthopaedic Centre, Windmill Road, Oxford OX3 7LD, UK

Thank you for your correspondence asking for us to specifically name the institution that classifies our work as a service evaluation and therefore allows us an ethical exemption. We have added this, as suggested, to line 134 of the manuscript. The revised manuscript has now been uploaded.

We look forward to hearing from you as to whether it may now be acceptable for publication.

Yours faithfully,

Angela Minassian (on behalf of the co-authors)